Author's response to reviews

Title: Are hygiene and public health interventions likely to improve outcomes for Australian Aboriginal children living in remote communities? A systematic review of the literature

Authors:

   Elizabeth L McDonald (liz.mcdonald@menzies.edu.au)
   Ross S Bailie (ross.bailie@menzies.edu.au)
   Brewster David (d.brewster@fsm.ac.fj)
   Peter Morris (peter.morris@menzies.edu.au)

Version: 2

Date: 22 February 2008